Y.2 Zeitschriften-Artikel

[1] Paasche, I. ” Aufgabe 714.“ Elem. Math. 30, 41–42, 1975.

[2] Erdös, P. ” Problem 3740.“ Amer. Math. Monthly 42, 396, 1935.

[3] Mordell, L. J., Barrow, D. F. ”Solution to Problem 3740.“ Amer. Math. Monthly 44, 252–254, 1937.

[4] Bankoff, L. ” An Elementary Proof of the Erdös-Mordell Theorem“ Amer. Math.

Monthly 65, 521, 1958. [5] Hungerbühler, N. ” A short elementary proof of the Mohr-Mascheroni theorem“ Amer. Math. Monthly 101, 784–787, 1994.

[6] Shapiro, H. S. ”Problem 4603.“ Amer. Math. Monthly 61, 571, 1954.

[7] Troesch, B. A. ” The Validity of Shapiro’s Cyclic Inequality“ Math. Comput. 53, 657–664, 1989.

[8] Clausing, A. ”A review of Shapiro’s cyclic inequality“, in [Wal92], S. 17–31.